Author’s response to reviews

Title: The association between delusional-like experiences, and tobacco, alcohol or cannabis use: a nationwide population-based survey.

Authors:

Sukanta Saha (sukanta_saha@qcmhr.uq.edu.au)
James G Scott (James_G_Scott@health.qld.gov.au)
Daniel Varghese (Daniel_Varghese@health.qld.gov.au)
Louisa Degenhardt (louisa@burnet.edu.au)
Tim Slade (tims@unsw.edu.au)
John J McGrath (John_mcgrath@qcmhr.uq.edu.au)

Version: 3 Date: 12 October 2011

Author’s response to reviews:

Dear Colleague,

We have adjusted to abstract heading to comply with BMC formats, and added text about response rates (see below).

Kind regards,

John McGrath

"As the response rate for this survey was lower than expected (60%), extensive non-response analyses were undertaken by the Australian Bureau of Statistics in order to assess the reliability of the survey estimates. As a result, adjustments were made to the weighting strategy (full details available elsewhere http://www.abs.gov.au [23])."